Author's response to reviews

Title: Impact of a web-assisted tobacco quality improvement intervention of subsequent smoker behavior: A DPBRN Study

Authors:

Thomas K Houston II (thomas.houston@umassmed.edu)
Kathryn L DeLaughter (kathryn.delaughter@umassmed.edu)
Midge N. Ray (midgeray@uab.edu)
Gregg H. Gilbert (ghg@umassmed.edu)
Jeroan J. Allison (jeroan.allison@umassmed.edu)
Catarina I. Kiefe (catarina.kiefe@umassmed.edu)
Julie E. Volkman (julie.volkman@va.gov)

Version: 2 Date: 6 June 2012

Author's response to reviews: see over
June 6, 2012

BMC Oral Health
BioMed Central
Floor 6, 236 Gray’s Inn Road
London
WC1X 8HB
United Kingdom

RE: Change in Title of manuscript

Dear Editors,

Please note the change in the title of the submitted manuscript from: “Oralcancerprevention.org: Impact of a web-assisted tobacco quality improvement intervention of subsequent smoker behavior” to: “Impact of a web-assisted tobacco quality improvement intervention of subsequent smoker behavior: A DPBRN Study”

Please advise if you need additional information. Thank you for your consideration.

Sincerely,

Thomas Houston, MD, MPH
Professor and Chief Division of Health Informatics
University of Massachusetts Medical School
55 Lake Avenue North
Worcester, MA 01655